Author’s response to reviews

**Title:** A protein interaction map identifies existing drugs targeting SARS-CoV-2

**Authors:**
claudia cava (claudia.cava@ibfm.cnr.it)
Gloria Bertoli (gloria.bertoli@ibfm.cnr.it)
Isabella Castiglioni (isabella.castiglioni@ibfm.cnr.it)

**Version:** 4  **Date:** 11 Aug 2020

**Author’s response to reviews:**

1. Your abstract should include the following headings: Background, Methods, Results, Conclusions, and should be no longer than 350 words. We note a Methods heading is missing. Please ensure you apply this format to the abstract in the main text and the submission system also.

   Thank you. Done

2. In order to be in line with journal requirements, please ensure the headings within your manuscript are labeled the following: Background, Methods, Results, Discussion, Conclusions.

   Thank you. Done

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

   Thank you. Done